	1	TAFF JOURNAL OR DUTY OFFICER'S LOG	
	d Bn (Abn), 50	FROM TO	
ITEM	TIME	UUUI 14 FED COM 2400 14 FE	INI-
NO.	IN OUT	INCIDENTS, MESSAGES, ORDERS, ETC.	TIAL
1	0050	Fr @sst G3 to W3: The 2 Quad 50 have been	
		coordinated with Div Arty. Coord directly	
		with Div Arty as to the t ime and plade	-
		H ave not received trucks for today from	-
		G4 hopefully will have by 0730 will contact	
		you. Direct coord with provost marshall	
		authouity for convoy to P K 17	
2	0100	SITREP Negative J	
3	0125	SITREP Negative A, B, C, 10, E, J	
4	0200	SITREP Negative	
5	0225	SITREP Negative A, B, C, 10E J	
6	0300	SITREP Negative	
7	0325	SITREP Negative	1
8	0400	SITREP Negative	
9	0425	SITREP Negative	
10	0500	SITREP Negative	
11	0525	SITREP Negative	
12	0615	SITREP Negative	
13	0625	SITREP Negative	
14	0705	SITREP Negative	
15	0721	MS team leaving this loo now.	
16	0727	SITREP Negative	
17	0745	MS team at checkpoint 101	

	DAILYS	STAFF JOURNAL OR DUTY OFFICER'S LOG (AR 220-346)	PAGE NO.	NO. OF PAGES
ORGANI	ZATION OR INSTALL	LOCATION	PERIOD COVER	ED
		FROM ···		то
		HOUR DATE	HOUR	DATE
ITEM NO.	TIME IN OUT	UINCIDENTS, MESSAGES, ORDERS, ETC.	ACTION T	AKEN INI-
18	0910	Fr D Co: Still only 2 lig htlywounded, sti	11	
		need some pickup and medical supply.		
		Request he already be sen t.	J 00	
19	0817	Clis now moving		
20	0820	4.2 inch mortar, fire 89HE, 322 HEleft		
		243111 left, 0 WP, 0F.S.1 eft		
21	0831	C 2 is now moving		
22	0836	A Co moving at this time		
23	0839	Rcn is now moving		
24	0842	A 16 MS team at CP 106		
25	0842	Call From Salvo 3 Mike Cpt "illiams:		
		He will spot quad 50 where and when		
		desire call back and let him know only		
		one available for today.		
26	0853	C 3 is now moving		
27	0907	Co A report about 30 pers moving over the	States and states	
		hill side at 523297 or 8They belong to C		
28	0908	C 6 and A 6 coord with each other so the	У	
		both know where both elem ens are		
29	0913	C 6 rpts C 1 at patrol base, C2 from SOW	and the second state of th	
		RD 3.8 1.8 YD 528292, C3 From FLOOR LD 4.5 1.5 YD 545295		
30	091/	Ren fm floow ID 3.8 0.2 XD 552308	age avec seen also success and the set	

RGANIZ	ATION OR I	NSTALLA	TION LOCATION PERIOD COVER	RED
			FROM	то
			HOUR DATE	DATE
NO.		OUT	ACTION	TAKEN INI-
31	0918		A 16 at CP 102	
32	0922		A Co loc YD 520300	
33	0930		Rcn position for remainde r of day YD 552308	
34	0935	2	C 1,2,3, have closed on P B	
35	1001		A 16 at CP 103	
36	1004		Cl an d C3 depart on a mission	-
37	1022		A Co from SEE D 1.8 YD 490292	
38	1037		A 16 CP 107	
39	1053		C3 from SEW RD 5.2 3.7 YD 542273	
			C 1 from SEW RD 3.0 3.5 Y D 520275	
	-		C 2 at PB	
40	1054		A 16 linked up with team coming from the	
		-	south returning to this loc ation.	
41	1107		Fr A Co: fr SEW LD 0.5 1. 5 YD 485295	
47	1128		Msg fm 6 to A6: He back no later than 1500	
48	1145		Cl found cash at coord 520270. Kakhi	
			uniform and civilian shirt and a VC	
			bandner with VC writing o n it.	
49	1345		Fr C Co: element locations	
			C 1 SEW RD 3.0 3.5 YD 520275	knardin j
			C 2 Fr FLOWER LD 4.0 3.8550272	
			C 3 at company patrol Base .	

RGANIZ	ATION OR INS	TALLATION	LOCATION		PERI	OD COVER	RED	
				FROM	M		то	
				HOUR DATE	E Transition des considerat	HOUR	DATE	
			ALLAL		0			
ITEM NO.	TIME	out U D	THEIDENTS, MESSAGES, O	RDERS, ETC.	1	ACTION .	TAKEN	INI- TIAL
-		and the second s	nter andere mandel over mandel and anti-anti-anti-anti-anti-anti-anti-anti-	an a	10-10-00-00-00-00-00-00-00-00-00-00-00-0	9		
51	1315	Fm A	Co location 500296					
50	1000							
52	1352	Pm	C Co Cl at patrol bas					
53	1400	C3	preparing to move out	PB				
-	1105		VD FO/000					
54	1405	62.	location YD 526290					-
55	1409	LRRP	30 from SEW RD 8.9 3	.6 YD				
FL.	4148	D. A	0. 1					
56	1415	Fm A	Co lead element clos:	ing in this loc				
57	1420	Fm C	Co Clleaving PB 1	420				
FO	1126		0					
58	1436	Fm A	Co: all elements clos	sed				-
59	1505	C Cc	o elem has relieved B	Co at WP		(
60	1515	0.24	C 2 closed nonimeter					
00	1212	020	& C 3 closed perimete	r				-
61	1520	010	closed perimeter					
62	1600	Fm F	B Co: first elem now m	oving				
	1000		JOS. 11150 CLOM NOW I	IO VING				
63	1635	B11	Located 545301					
64	1700	Fm B	n XO to CO: The move	to Bein Hos				
~~	1100	- 141 L	H 10 00 cc. The move	to bein noa	-			
		will	begin on 16 Feb 68.	The order of				
		mover	nent will be 1/501, 2/	/501 Bde				
		eleme	ent 2/502. We do not	know how long				
		the m	novement will take. At	present time				
				process vinc				
		we do	not know the availab	oility of trans				
65	1700	Fr 3d	Bde: We are moving h	ack at this tir	ne			
		Anagerican server	To serve a survey survey and a survey with a second sound the Agram of Demonstration and the second sound sound					
		ab ou	at 500 m from the brid	ge, we had only	y	single realization and the second		
		+	sualty. He was KIA. 1	FILL	411			

			(AR 220-346)		C	16	
ORGANIZ	ATION OR	INSTALLA	TION DOCATION PI	ERIOD	COVER	TO	
			HOURE DATE	NAME OF TAXABLE	HOUR	DATE	
			A THE WALL REAL PROPERTY AND A THE		1		
ITEM	TI	ME	INCIDENTS, MESSAGES, ORDERS, ETC.	ixo A	CTION .	TAKEN	INI-
NO.	IN	OUT		1	ł.		TIAL
66	1700		To 1 ACD secure unit locations and Co B				
			night ambush sites				
67	1745		Co B loc B1, 562302; B2 558308				
			B3 & CP 548308				
68	1820		Rc n plt moving this time				
			a to be to state at 1000 to enhance atta				
69	1822		Co A plt departs at 1820 to ambush site				
70	1835		Ren located 558319				-
~	1015		Received overlay from 1/7 Cav we have				
71	1845		Received overlay from 1/7 day we have				+
			granted A 1/7 permsiiion to operate in				
			our AO as indicated on overlay				
				-			
72	1925		Co D during Sd operation vic of 631310				
			found 3 Ml rifles, 5 carbines, 12 rds 60mm				
			mort, 4 magazines .30cal ammo misc web ge	ar.			
73	1930		SITREP all units negative				
74	1934		Co A loc 528300 will set up this loc for n	igne.			-
75	1945		Co B loc C3 & CP 551307, C2 558307, C1 5703	302			
76	19	6	Rcn located 560324				
/0	17	40					+
77	1947		Co B HI M 79 rds impacti ng 300-400from 55	307			
78	2010		To 1 ACD secure night ambu sh location SITRE	Neg			
	2010						
79	2020		Ren 1 oc 577304	*			
80	2020	2025	SI TREP Negative	8	1		
81	2043		Rcn loc 577313 will remain here tonight				

		DAILY ST	AFF JOURNAL OR DUTY OFFICER'S LOG (AR 220-346)	PA	GE NO.	NO. OF PAGE	ES
ORGANI	ZATION OR	INSTALLAT	LOCATION	PER	IOD COVE	RED	
		1	THE REPORT OF THE PARTY OF THE	OM		то	
		All solutions and a solution of the	CONFIDENTON	T Bint Aurentin	HOUR	DATE	
ITEM NO.	IN	ME -	INCIDENTS, MESSAGES, ORDERS, ETC.	See .	ACTION	TAKEN TI	IALS
83	2200		SITREP Negative	HEREIN MICH. MANNEN MINING	none, 22		
84	2216		Fr D TOC : have to provi de 2 sods to				
			accompany EOD team outside the perimete	er			
	•		between 1300 1500 hrs. Contact Sgt				
			Holchecker Sky King 204 a pprox 0900 hr				and a second
85	22	25	SITREP Negative				
86	2300		SITREP Negative				
87	2325	2330	SITREP Negative				
88	2400		closed out				
			and a state of the state of t				
			L'A ANTIMEN	-			
					i		
			antenin di alta de alta de antenin terra puesta nan menangin unternet terra a da antenin de antenin de antenin d	an average of antis	and manipal of the single types of	and the state of the	

